Author’s response to reviews

Title: Association of Functional and Structural Social Support with Chronic Kidney Disease among African Americans: The Jackson Heart Study

Authors:

Rasheeda Hall (rasheeda.stephens@duke.edu)

Clemotina Davenport (tina.davenport@duke.edu)

Mario Sims (msims2@umc.edu)

Cathleen Colon-Emeric (cathleen.colonemeric@duke.edu)

Tiffany Washington (twashing@uga.edu)

Jennifer St. Clair Russell (jenstclrus@gmail.com)

Jane Pendergast (jane.pendergast@duke.edu)

Nrupen Bhavsar (nrupen.bhavsar@duke.edu)

Julia Scialla (julia.scialla@duke.edu)

Crystal Tyson (crystal.simpson@duke.edu)

Wei Wang (souwwang80@gmail.com)

Nancy Min (ymin@umc.edu)

Bessie Young (youngb@u.washington.edu)

L. Boulware (ebony.boulware@duke.edu)

Clarissa Diamantidis (clarissa.diamantidis@duke.edu)

Version: 1 Date: 23 May 2019

Author’s response to reviews:

Our cover letter includes a table that describes our point-by-point response to each editor and reviewer comment. Thank you for considering this revised manuscript.